Author’s response to reviews

Title: Surgical therapy of primary hepatic angiosarcoma

Authors:

Verena Tripke (verena.tripke@unimedizin-mainz.de)
Stefan Heinrich (stefan.heinrich@unimedizin-mainz.de)
Tobias Huber (tobias.huber@unimedizin-mainz.de)
Jens Mittler (jens.mittler@unimedizin-mainz.de)
Maria Hoppe-Lotichius (maria.hoppe-lotichius@unimedizin-mainz.de)
Beate Straub (beate.straub@unimedizin-mainz.de)
Hauke Lang (hauke.lang@unimedizin-mainz.de)

Version: 1 Date: 13 Nov 2018

Author’s response to reviews:

1. Ethical approval and consent to participate: Ethical approval not applicable. Patients agree that their medical data can be used anonymized for retrospective studies when signing the treatment contract.

2. Figures: Larger Version of the figures included